Author's response to reviews

Title: The expression of type 1 plasminogen activator inhibitor (PAI-1) in clear cell renal cell carcinoma (CCRCC) and its impact on angiogenesis, progression and patient survival after radical nephrectomy

Authors:

Dragomir P Zubac (dragomir.zubac@kir.uib.no)
Tore Wentzel-Larsen (tore.wentzellarsen@gmail.com)
Tomas Seidal (tomas.seidal@lthalland.se)
Leif Bostad (leif.bostad@helse-bergen.no)

Version: 5 Date: 28 November 2010

Author's response to reviews:

Dear Editor,

We changed subtitel Methods to Materials and methods in the Abstract. Terms like Fig. and Table have been changed to Figure or Table accordingly with arabic numerals consecutively throughout the manuscript. The margines to Figures 1 and 2 have been reduced now.